Author’s response to reviews

Title: Social Determinants of Health Associated with Hemodialysis Non-Adherence and Emergency Department Utilization: A Pilot Observational Study

Authors:

Kamna Singh Balhara (kamna.balhara@gmail.com)
Lori Fisher (lori2k15@gmail.com)
Naya El-Hage (naya.elhage@ahn.org)
Rosemarie G Ramos (rosemarieramos@hotmail.com)
Bernard G Jaar (bjaar@jhmi.edu)

Version: 2 Date: 07 Dec 2019

Author’s response to reviews:

***We would like to thank the editorial board and reviewers for their thoughtful consideration of our manuscript.***  
Assistant Editor Comments: 1. We note author Bernard is a member of the editorial board. To ensure transparency, please declare this in the Competing Interests section of the Declarations. ***This has been updated accordingly.***  
2. Please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter. Please proofread and ensure that when you upload your revised submission it is in the final form for publication. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Additional files should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.  
Associate Editor Comments: We hope to see a follow-up study on addressing this issues if you are able to demonstrate intervention(s) that help improve compliance and lessen E.R. visits and unnecessary hospitalizations *** Thank you; it is our hope that this pilot study will inform the development and evaluation of targeted interventions to decrease non-adherence or mitigate its impacts.***